Author's response to reviews

Title: Differential in vivo expression of mycobacterial antigens in Mycobacterium tuberculosis infected lungs and lymph node tissues

Authors:

Tehmina Mustafa (tehmina.mustafa@cih.uib.no)
Nils Anders Leversen (nilsleversen@fastmail.fm)
Lisbet Sviland (lisbet.sviland@helse-bergen.no)
Harald Gotten Wiker (harald.wiker@k2.uib.no)

Version: 2 Date: 9 May 2014

Author's response to reviews:

Following changes are made in the manuscript:

1. Line numbers are added.

2. Ethical statement and consent statement included in the manuscript. Ethical clearance was obtained from the regional ethical committee of western Norway. There was no written informed consent as the biopsies were obtained from the archives. The ethical committee was informed about the difficulties in obtaining an informed written consent and the project was cleared.